Author's response to reviews

Title: Expression of tumor-specific antigen MAGE, GAGE and BAGE in ovarian cancer tissues and cell lines

Authors:

  Shiqian Zhang (zhangshiqian370112@126.com)
  Xiaoliang Zhou (zhou3208@163.com)
  Hao Yu (yuhao19@yahoo.cn)
  Yunhai Yu (yyhwdmlove@yahoo.com.cn)

Version: 6 Date: 19 April 2010

Author's response to reviews: see over
Dear respected editors,

We appreciate your careful reminders regarding the formatting changes for the original manuscript. In this version, we have carefully addressed all points raised by the editors. Our specific responses to each of the points are summarized on the following response sheet.

In summary, the manuscript has been thoroughly revised. Thank you so much for your acceptance for publication.

Structure: Please insert a heading titled 'Conclusions' after the discussion. Alternately, please rename the discussion 'Discussion and conclusions'.

Re: thank you! We've revised it as you suggested.

Email address: please use the following format for email addresses, including using authors' initials: AB: abcd@institution.ac.uk EF: efgh@generic.co.uk IJ: IJKL@corporation.com

Re: thank you! We've revised the format for email addresses as you requested.

Abstract: Please remove the keywords from the abstract.

Re: thank you! We’ve deleted the keywords from the abstract section.

Text: Please ensure that all text is presented in black.
Re: thank you! We’ve checked the text color.

Underlining: Please note that underlining is not supported in the final full text; you may wish to change it to bold.

Re: thank you! We’ve corrected it.

Materials and methods: Please rename this section 'Methods'.

Re: thank you! We’ve renamed the section as you requested.

Acknowledgements: Please insert this section after the authors' contributions.

Re: thank you! We’ve adjusted the position of Acknowledgements section.

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements.

Re: thank you for your careful reminder! We’ve re-designed the table layout.